Author's response to reviews

Title: Management of neonatal sepsis at Muhimbili National Hospital in Dar es Salaam: diagnostic accuracy of C-reactive protein and newborn scale of sepsis and antimicrobial resistance pattern of etiological bacteria

Authors:

Martha F Mkony (mmkony@gmail.com)
Mucho M Mizinduko (mucho_m2002@yahoo.co.uk)
Augustine Massawe (dr.augustine.massawe@gmail.com)
Mecky Matee (mateemecky@yahoo.com)

Version: 4 Date: 1 November 2014

Author's response to reviews:

The following has been addressed as per reviewers comments

1. copy editing the document: done
2. to provide single email for each other: provided
3. figure tile should include numbers: edited
4. remove the word appendix from the document: word removed
5. remove duplicated figures: figures removed from the main document.